Author’s response to reviews
Title: Acute and Sub-Acute Oral Toxicity of Dracaena Cinnabari Resin Methanol Extract in Rats

Authors:

Nashwan Al-Afifi (dr_nash1@yahoo.com)
Aied Alabsi (aied_absi@yahoo.com)
Marina Bakri (marinab@um.edu.my)
Anand Ramanathan (drranand@um.edu.my)

Version: 2 Date: 30 Dec 2017

Author’s response to reviews:

Comment

1 Please move all figure legends the end of the main manuscript in a section labelled, "Figure legends".

Response: Done. (Figure legends section, line 666-674. page 32).

2 Please revise your Authors' contribution section to confirm all authors have read and approved of the final manuscript. Additionally, please refer to all authors by their initials in this section.

Response: Done.

3 Please delete all of your supplementary files. They are no longer necessary at this point.

Response: Done.
Thank you for including a data availability statement. This statement should detail the location of the raw data underlying the conclusions made in this study. We note that only summarized data is presented in this manuscript, so we would like to ask that you amend this statement. If there are any ethics restrictions preventing the sharing of the raw data, please state this. If you do not wish to share the data, please include a reason why it will not be shared in this section.

Response:

Done. All data generated or analysed during this study are included in this published article [and its supplementary information files].